STUDY PROTOCOL

Transitioning to digital first line intervention – validation of a brief online screener for early identification of a suspected eating disorder: study protocol

Emma Bryant^{1*}, Jane Miskovic-Wheatley², Stephen Touyz², Ross D. Crosby³, Eyza Koreshe², Li Cao³ and Sarah Maguire²

Abstract

Background: Only one in four people with eating disorders seeks treatment, and of those who do seek treatment, 20% go on to experience a chronic course. Early intervention has been associated with better prognosis, with those seeking specialised intervention in the early stages of their illness more than twice as likely to achieve remission. Current screening measures typically require expert administration and are rarely validated across a spectrum of DSM-5 eating disorder presentations or for online use. In light of COVID-19 and increasing reliance on telehealth technologies in the intervention and delivery of mental health services, online assessments suitable for self-referral are likely to be the first step to seeking care. InsideOut Institute has developed a 6-item online screening tool for the purposes of identifying eating disorder risk and symptomatology, aimed specifically at increasing help-seeking behaviour in subsyndromal and early presentations.

Methods: This study investigates the reliability and validity of the InsideOut Institute Screener (IOI-S), using a cross-sectional survey research design. Participants aged 14 and over will complete an extensive baseline survey battery for evaluation. 50% of participants will be randomly selected for one follow-up re-test of the IOI-S only, 2 weeks post initial testing. The IOI-S will be analysed for statistical reliability on two parameters: internal consistency and test re-test reliability, and for statistical validity on four parameters: concurrent validity, sensitivity and specificity, convergent and discriminant validity.

Discussion: The rapid and ongoing shift to digital intervention has highlighted gaps and opportunities in our pathways to care. Adequate screening for eating disorders is a major gap. This study aims to validate an online screening tool for use in telehealth early intervention, designed for users seeking information for a suspected eating disorder. The screener meets those at risk 'where they are' (i.e. online) and may improve timely referrals to relevant services. This is of particular salience as face-to-face healthcare and traditional frontline interventions are disrupted, and we are challenged to re-design our practices to deliver diagnostic and treatment services in highly adaptive digital contexts.

Keywords: Eating disorders, Early intervention, COVID-19, Psychometrics, Telehealth, Digital, Screening questionnaire

* Correspondence: ebry8657@uni.sydney.edu.au

¹School of Medical Sciences, Faculty of Medicine and Health, The University of Sydney, Sydney, Australia



[©] The Author(s). 2020 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.



Open Access

Full list of author information is available at the end of the article

Plain English summary

Early intervention greatly improves treatment outcome, however currently only 1 in 4 people with an eating disorder seeks treatment. Current screening measures are typically designed to identify traditional cases of Anorexia and Bulimia Nervosa rather than address a spectrum of eating disorder behaviour and are rarely validated for online use. Adequate screening designed to encourage help-seeking behaviour is needed. Given 84% of Australians now seek healthcare information online before going anywhere else, and with a rapid COVID-19 driven shift to digital healthcare and intervention, novel online screening tools designed to identify broad eating disorder risk and symptomatology may become the first step to seeking care. This study aims to validate a 6-item online screening tool developed by the InsideOut Institute for Eating Disorders at the University of Sydney in a diverse population of people aged 14 and over. It is designed to 'start a conversation' so that it may be used to encourage help-seeking behaviour and improve timely referrals to relevant services (digital or otherwise).

Background

Eating Disorders (ED), marked by persistent disturbed eating behaviour and negative psychological affect, are complex psychiatric conditions associated with life-threatening physical complications [1] and high concurrent mood disorders [2, 3]. Anorexia (AN) and Bulimia Nervosa (BN) alone cause 1.9 million disability-adjusted life years globally [4] and cost the Australian economy \$69.7 billion per year [5]. Despite this, only one in four people with EDs seek treatment [6], and of those who seek treatment, 20 % will become chronic [7]. Elevated mortality rates and relatively poor treatment outcomes may in part be attributable to a paucity in help-seeking and the frequent delay between illness onset and intervention – a factor known to heavily influence prognosis [8, 9].

Early intervention is paramount to improved treatment outcome [10, 11], however is frequently impeded by mental health illiteracy, self-stigma and shame [12]. Eating disorder stereotypes exacerbate ambivalence towards treatment seeking [13] and contribute to the ego-syntonic nature of the illness, where people can be stuck between the precontemplation (denial) and contemplation (awareness with resistance) stages as described by Prochaska's transtheoretical model of change [14]. This ambivalence may be compounded by cultural normalisation of dietary and exercise excess, contributing to confusion around experiences of pathology and distress [15].

Screening tools are regularly used to triage medical and psychological risk in healthcare settings and have been shown to be effective in early identification, often triggering referral and further assessment where results suggest a condition may be present [16]. They may have superior utility over long form diagnostic tools in the crucial pre-contemplative stage of an eating disorder, mitigating the challenges clients frequently face in being properly identified by time-poor primary healthcare physicians, many of whom lack the expertise or feel ill-equipped to deliver complex eating disorder assessments [17, 18] such as the gold standard Eating Disorder Examination Questionnaire [19, 20].

A number of screener questionnaires have been validated for use in eating disorders. These include the widely used 5-item Sick, Control, One Stone, Fat, Food (SCOFF) questionnaire (Morgan, Reid & Lacey, [21]) and the 5-item Eating Disorder Screen for Primary Care (ESP) (Cotton, Ball & Robinson, [22]). The SCOFF is brief and has demonstrated good internal consistency and concurrent validity with the EDE-Q, however, shows limited evidence identifying disordered behaviour beyond traditional presentations of AN and BN [23]. Additionally, language may assume acceptance of an eating disorder diagnosis, thus may not be useful in the prodromal phase of the illness. The SCOFF has attracted criticism for its use of confronting behaviour-based questions (e.g. 'Do you make yourself sick?', 'Have you lost more than one stone?') [23]. These types of questions have been found to be prone to denial or distortion on face-to-face assessment [24]. Moreover, the screener has limited specificity-the authors agreed maximising sensitivity was paramount given the morbidity and mortality associated with eating disorders and the consequences of a false negative result, thus reducing the cutoff from three positive answers to two [21]. Importantly, as with most existing screeners, the SCOFF is validated for in-person use only, and thus assumes a person is already in the care of a healthcare professional at the time of taking the test.

The Eating Disorder Screen for Primary Care is more sensitive but less specific than SCOFF at detecting eating disorders [25]. It employs more sympathetic language and includes questions about personal and family history of eating disorder diagnoses, however the latter has been found to have no predictive strength and is dropped from analyses in many cases [25]. The ESP has not been widely circulated or validated.

Several short forms of the EDE-Q, including the EDE-QS (Gideon et al., [26]) (12 items), the EDE-Q7 (Grilo, Henderon, Bell & Crosby, [27]) and the EDE-Q8 (Kliem et al., [28]) have been conceived and psychometrically validated across populations, demonstrating good sensitivity and internal consistency and adequately correlating with the original EDE-Q. However, all utilise the original items, derived from previous iterations of the DSM and prior to the inclusion of diagnostic categories Binge Eating Disorder (BED), Avoidant Restrictive Food Intake Disorder (ARFID), Other Specified Feeding or Eating

Disorder (OSFED) and Atypical AN, and again, are not specifically designed for online use.

To our knowledge, there have been few intermediary measures and novel screening tools developed since the 2013 revision of the DSM (5) despite changes to existing diagnostic categories and the inclusion of a BED diagnosis. Further, existing diagnostic and screening measures are typically validated to discriminate between healthy and ill, rather than using a continuum to identify those showing early symptomatology/risk [29]. One recent exception is the Stanford-Washington University Eating Disorder screen (SWED) (Graham et al., [30]) which assigns a risk threshold and has been validated for online use [30, 31]. However, this questionnaire is 17 items long and has been primarily validated in college age females. Given the prognostic implications of early intervention, there is evidenced need to expand on current validated screeners for diverse populations, and to do so in a way that reflects modern-day healthcare and the move to online health information-seeking [32].

Healthcare is experiencing a paradigmatic shift from medical paternalism to autonomous patient-centred care [33] with people increasingly turning to internet technologies for support and advice [34]. Eighty-four percent of Australians log on to the internet to search healthcare information before going anywhere else [35]. The most at-risk population for eating disorders, young people, are significantly more likely to seek initial information online than to visit their GP [36]. The occasion of the COVID-19 global pandemic has driven a sudden shift to online healthcare delivery [37], alongside increased rates of mental health complaints and worsening anxiety and depression [38, 39]. Those suffering from eating disorders are particularly vulnerable during times of food insecurity, with lack of access to "safe" foods combined with erratic food distribution causing a 'feast or famine' pattern defined by hoarding, restriction, bingeing and compensation [40, 41]. There is evident emergent need, then, for novel online risk-assessments and screening options suitable for self-referral, where the burden of mental health is increasing while face-to-face access to physicians is curtailed by social distancing measures [42, 43]. New screening tools should address the semantic and methodological limitations of present standardised measures and be validated for online use across the spectrum of eating disorders as defined by the DSM-5 (including presentations of BED, ARFID and OSFED). They may be more acceptable and less susceptible to denial and distortion by employing non-confrontational language [24]. To this end, InsideOut Institute for Eating Disorders (IOI) within the Boden Collaboration for Obesity, Nutrition, Exercise and Eating Disorders, The University of Sydney, conceptualised a short, 6-item online screening tool, the Inside-Out Institute Screener (IOI-S, [44]) to assess for eating disorder symptomatology/risk using non-intrusive, sensitive language designed to 'start a conversation.'

This study aims to psychometrically validate the IOI online screener so that it may be used reliably as an adjunct or alternative to existing screening measures and/ or for referral purposes, encouraging online users who may otherwise have delayed seeking professional help to do so with self-governance and autonomy.

Methods/design

Participants

Approximately 250 participants will be recruited into the study. They will be aged 14 and over, of all genders, and divided into two groups – clinical and non-clinical. Clinical status will be determined post-testing using EDE-Q cut-off scores. We aim to recruit across all DSM 5 diagnostic categories for eating disorders.

Participants will be recruited in two ways:

- 1. *Amazon's Mechanical Turk (MTurk):* 'MTurk' is a global internet crowdsourcing service used frequently in academic research to recruit a broad base of willing research participants who are compensated for completing surveys, questionnaires and tasks as assigned by the researcher [45]. MTurk does not permit people under 18 to use its service: thus, it will be used to recruit adults aged 18+ only, both clinical and non-clinical. Participants will be able to sign up from anywhere in the world, deeming it an international cohort. They will be reimbursed according to MTurk recommendations.
- 2. *Community:* Participants will be recruited via online advertisements on health websites and social media channels/paper advertisements. Advertisements will be aimed at individuals who self-report an eating disorder diagnosis, though those without an eating disorder are also welcome to participate, and all will go into the draw to win a Westfield gift card. We are specifically looking to recruit adolescents via this method, due to MTurk's restriction on underage participants, however adults will also be targeted.

The use of two methods of recruitment aims to increase generalisability across age, ethnicity, gender and Body Mass Index (all of which will be reported) – this is significant due to the broad-reaching nature of the internet and the importance of validating the questionnaire across populations.

Inclusion/exclusion criteria

At this stage of validation, the screener will be conducted in the English language only, thus proficiency in reading the English language is required. Participants under the age of 14 are excluded from the study – separate validation studies are indicated for younger age groups due to differences in comparison measures, linguistic ability and potential impact on item comprehension. Also excluded are those with no reliable access to the internet. Otherwise, no further demographic or exclusion criteria will be set. Participants with a current diagnosis of an eating disorder as well as those with no current or former diagnosis are equally welcome to participate: these two groups (clinical and non-clinical) will be defined and compared post testing using EDE-Q cut-off scores.

Overall study design

This study will use a cross-sectional survey research design with follow up on a sub-sample to examine the 6item IOI-S for statistical reliability and validity. Participants will be given access to a baseline survey package delivered on the secure web survey platform REDCap. Approximately half the participants will be randomly selected to participate in a follow-up re-test involving completion of the IOI-S alone a second time 2 weeks post initial testing, to account for potential drop-out/uptake refusal. 30% is needed for sufficient test-retest reliability.

Measures

Four questionnaires make up the baseline survey package: The EDE-Q, the SCOFF Questionnaire, the IOI-S and either the Marlowe-Crowne Social Desirability Scale (MC-SDS) (adults) or Children's Social Desirability Short Scale (CSD-S) (adolescents), described in full below. The four existing surveys against which the IOI-S will be measured were chosen as the relevant diagnostic tools most frequently used in clinical settings and research trials. These will be presented in randomised order to the participants.

Baseline package

InsideOut institute screener (IOI-S) (InsideOut Institute for Eating Disorders, [44])

The six items of the new scale are displayed in Table 1. Item development followed review of the scientific literature for existing instruments screening and assessing eating disorder symptomatology, lived experience and clinical and research expert consultation. Instruments reviewed included the EDE-Q, the Eating Disorders Inventory (Garner et al., [46]), the SCOFF questionnaire, the ESP, and the Eating Attitudes Test (EAT; Garner et al., [47]), from which researchers developed an initial pool of 10 questions covering six facets of eating pathology: an individual's relationship with food, body, the extent to which body weight and shape determines selfworth, loss of control over eating, binge eating and compensatory behaviour. This was further narrowed by an expert consultation team to six relevant items to evaluate eating pathology and eating disorder risk in a non-clinical population. The IOI-S is rated on a 5-point Likert scale, where 1 is "never" and 5 is "all the time"; except for Question 1, where 1 is "worry and stress free" and 5 is "full of worry and stress". Responses are summed to yield a score between 6 and 30 points total, where 6 points is the lowest degree of risk and 30 points the highest degree of risk. Those deemed to be of moderate to high risk are directed to InsideOut's database of trained professionals with expertise in eating disorders.

Eating disorder examination questionnaire (EDE-Q) (Fairburn & Beglin, [20])

The EDE-Q is a 28-item self-report version of the EDE structured clinical interview which is regarded as the 'gold-standard' diagnostic tool in eating disorders and has been validated in multiple trials [48]. Traditionally paper-based, it has been psychometrically validated for online delivery [49]. It includes additional measures of weight, height, and missed menstrual periods, the latter of which will be excluded from our study due to the DSM-5 removal of and consequent diagnostic irrelevance of missed menstruation in ED. It is comprised of a global score and four subscales: Restraint, Eating Concern, Weight Concern, and Shape Concern, and employs a 7-point forced-choice severity rating where 0 points is the lowest severity and 6 points is the highest severity.

SCOFF (sick, control, one stone, fat, food) questionnaire (Morgan et al., [21])

The SCOFF questionnaire is a short 5-item forced choice (true/false) screening tool designed to assess eating disorder symptomatology. A threshold of > 2 positive answers indicates a 'likely case' of AN or BN. It has shown good internal consistency and concurrent validity with the EDE-Q [50]. The SCOFF is included due to the frequency with which it has been employed in clinical and research settings.

Eating Disorder Examination Questionnaire 8 (EDE-Q8) (Kliem et al., [28]

The EDE-Q8 is an 8-item short form of the longer EDE-Q, designed as an abbreviated outcome measure that retains the original factor structure. It is highly correlated with the EDE-Q and demonstrates strong internal consistency, concurrent and convergent validity. Items are drawn directly from and are identical to respective items in the EDE-Q and will be extracted for independent analysis in this instance.

Marlowe-Crowne Social Desirability Scale (MC-SDS) (Crowne & Marlowe, [51])

The MC-SDS is a well-validated 33-item self-report questionnaire measuring social desirability in adults by

Table 1	6-item	insideout	institute	screener

Page	5	of	8
------	---	----	---

Item	1	2	3	4	5
1. How is your relationship with food?	Worry and stress-free	A bit problematic	Moderately problematic	Very problematic	Full of worry and stress
2. Does your weight, body or shape make you feel bad about yourself?	Never	A little bit	Sometimes	Quite a bit	All the time
3. Do you feel like food, weight or your body shape dominates your life?	Never	A little bit	Sometimes	Quite a bit	All the time
4. Do you feel anxious or distressed when you are not in control of your food?	Never	A little bit	Sometimes	Quite a bit	All the time
5. Do you ever feel like you will not be able to stop eating or have lost control around food?	Never	A little bit	Sometimes	Quite a bit	All the time
6. When you think you have eaten too much, do you do anything to make up for it?	Never	A little bit	Sometimes	Quite a bit	All the time

evaluating concern with social approval. Items are rated true/false and balanced for positive and negative wording. It is frequently used as a measure of discriminant validity in instrument design [52] and has shown good divergent validity with measures of psychopathology (anxiety, depression) related to eating disorders [53]. The MC-SDS is not validated for use in children: thus, an adapted Children's Social Desirability Scale (CSD-S) will apply for those participants aged under 18.

Children's Social Desirability Short Scale (CSD-S) (Baxter et al, [54])

The CSD-S is an abbreviated 14-item version of the gold-standard 48-item Children's Social Desirability scale developed by Crandall, Crandall and Katovsky in [55]. The original scale was modelled on the MC-SDS for adults and is validated for use in a child/adolescent population [54]. It will be assigned to our 14 to 17-year-old participants. The scale uses binary response (yes/no) and is regularly used for methodological validity to detect confounding from social desirability bias. The CSD-S has demonstrated adequate internal consistency and test-retest reliability and good external validity [56].

Follow-up package

InsideOut institute screener (IOI-S) (InsideOut Institute for Eating Disorders, [44])

Follow-up involves completion of the 6-item IOI-S only.

Sample size calculation

Estimated sample size is based on 2.5–3 times the minimum sample size required for power as the outcome measure (statistical validity) is incrementally increased with participant numbers. Therefore, we anticipate a total sample size of approximately 250 participants.

Approximately 30% of participants in each of the clinical and non-clinical groups (defined post T1 on EDE-Q cut-off score) will need to engage in the second task for sufficient validity. Estimated minimum sample size is based on the following analysis: in each group 10 participants are required for a power of 0.80 to detect an intraclass correlation coefficient of 0.70; 80 participants are required for sufficient power to detect a Spearman's correlation of 0.40 or higher; 80 participants are required for good power to detect an Area Under Curve of 0.70, and at least 50 participants for a correlation coefficient of r +/- 0.70 for convergent and discriminant validity. All power analyses were conducted using PASS version 15 software [57].

Data analyses

Reliability

Reliability is an important feature of survey efficacy indicating the overall replicability and internal stability of a measure [58]. Measure reliability of the IOI-S will be examined on two parameters: internal consistency and test-retest reliability. The internal consistency, that is, the degree to which the six items on the scale measure the same underlying dimension, will be calculated using a standardized Cronbach's alpha score, with a minimum acceptable threshold of > .80 [59]. An exploratory factor analysis will be performed on the IOI-S items using principal axis factoring and Promax oblique rotation to examine factor loading, with an item-total correlation threshold of >.40 [60, 61]. Correlations between successive iterations of the test (test-retest reliability) will be reviewed using a 2-way mixed effects model Intraclass correlation coefficient with absolute agreement and a minimum acceptable threshold of > 0.70 [62].

Criterion validity

The degree to which the results on the IOI-S are able to predict an outcome on established measures of eating disorder symptomatology (concurrent validity) [63] will be assessed as concordance between the IOI-S, the EDE-Q, and the SCOFF using a Spearman's product moment correlation. Following, an Area Under Curve (AUC) analysis of Receiver Operating Characteristic (ROC) curve will be performed as analysis of IOI-S sensitivity and specificity threshold where a high sensitivity indicates accurate identification of true positive cases and high specificity indicates accurate identification of true negative cases [64]. Area Under Curve dictates the probability of identifying true positives or negatives by chance, where a value of 0.50 would suggest the measure is no more effective than guessing [65]. AUC threshold will be set at > 0.70 with a 95% confidence interval.

Construct validity

Convergent validity, or the degree to which the IOI-S correlates with eating disorder constructs and variables on both the EDE-Q and the SCOFF [66] will be measured with a correlation coefficient threshold set at r > 0.70. A correlation coefficient will also be determined for discriminant validity between the IOI-S and MC-SDS/CSD-S. A correlation coefficient of +/- 0.70 indicates a strong linear relationship between two variables [67].

Missing and outlier values

REDCap online survey platform can be configured to eliminate missing values, otherwise submission of the survey battery is not possible. If data is not normally distributed, Spearman's Rank Order will be used instead of Pearson's correlation coefficient.

Discussion

The internet is changing the way we seek and deliver healthcare, and current circumstances brought on by the COVID-19 global pandemic have intensified the need for validated online interventions and services, particularly where appropriate intervention at first point of contact is shown to be of fundamental gain to outcome. The primary outcome of this study is to establish if the Inside Out Institute Screener is a statistically reliable, valid and accurate online measure of eating disorder risk and symptomatology in those aged 14 and over so that it can be reliably used by those seeking information for a suspected eating disorder. The development of the tool was predicated on the understanding people now pursue support and information for healthcare concerns on the internet before presenting anywhere else, and the opportunity this presents to inform and encourage helpseeking behaviour in a population largely going undetected in primary healthcare [18]. Although primarily intended for personal use and potential self-referral, the screener is further designed to accurately identify clinical status and thus has the potential to be used in a clinical setting to distinguish true cases from non-cases, with treatment elucidated on that basis.

Questionnaires used in the field of eating disorders have mostly not been validated for online use. Instead, scores on each measure and respective community/ clinical norms are derived from paper administration. This is often performed in conjunction with a trained clinician, which as stated, can confound results due to the nature of these types of questions being prone to denial and distortion on face-to-face assessment. Thus, lack of available norms for online administration may impact the findings. However, inclusion of the current gold-standard screening tools against which to validate was determined to be of necessary importance. Additionally, as all surveys are being delivered online in this instance, any effect due to test condition will be homogenous across the survey sample.

This study aims to recruit a large pool of applicants of diverse ethnicity, age, clinical presentation and gender. This is achieved by the use of two recruitment arms – one of which, MTurk, has a broad global reach – and will greatly increase generalisability. It will be one of the first to assess and identify risk on a spectrum rather than classifying true cases and non-cases only, with the aim of improving early intervention in subsyndromal presentations.

The study's significance will lie in its contribution to e-health early intervention and general screening efficacy across a spectrum of eating disorders. Given the prognostic implications of early intervention and current circumstances surrounding COVID-19, the importance of building e-health resources for an adaptable health system intended to encourage help-seeking cannot be underestimated.

Abbreviations

AN: Anorexia Nervosa; ARFID: Avoidant Restrictive Food Intake Disorder; BED: Binge Eating Disorder; BN: Bulimia Nervosa; EAT: Eating Attitudes Test; ED: Eating Disorder; EDE-Q: Eating Disorder Examination Questionnaire; EDE-QS: Eating Disorder Examination Questionnaire Short; EDI: Eating Disorders Inventory; IOI-S: InsideOut Institute Screener; OSFED: Other Specified Feeding or Eating Disorder; SCOFF: Sick, Control, One Stone, Fat Food questionnaire; MC-SDS: Marlowe-Crowne Social Desirability Scale; CSD-S: Children's Social Desirability Short Scale

Acknowledgements

We would like to thank Gabriella Barclay, Sarah Horsfield and Bronny Carroll for valued and expert assistance in development of the screening tool.

Authors' contributions

EB, JMW and SM designed the study protocol; RDC and LC assisted with power calculation and statistical analysis design; ST, EK, JMW and SM reviewed the study protocol; EB wrote the first draft; All authors edited, read and approved the final manuscript.

Funding

Research costs for this work are supported by the InsideOut Institute for Eating Disorders. There is no external funding.

Availability of data and materials

Not applicable

Ethics approval and consent to participate

This study is approved by the Human Research Ethics Committee (HREC) at the University of Sydney in compliance with the *University of Sydney Research Code of Conduct 2019* guidelines (2020–363). Prior to undertaking any study-related procedures, each participant will view an online information form

outlining study aims, procedures, potential risks and benefits and will provide electronic informed consent. Separate consent will be obtained from parents/quardians of participants under the age of 18.

Consent for publication

Not applicable.

Competing interests

Stephen Touyz is a founding co-editor in chief of the Journal of Eating Disorders and receives honoraria from Biomed Central.

Author details

¹School of Medical Sciences, Faculty of Medicine and Health, The University of Sydney, Sydney, Australia. ²InsideOut Institute for Eating Disorders, The Boden Collaboration for Obesity, Nutrition, Exercise and Eating Disorders, The University of Sydney, Sydney, Australia. ³Sanford Center for Biobehavioral Research, Sanford Research, Fargo, ND, USA.

Received: 30 July 2020 Accepted: 14 October 2020 Published online: 12 November 2020

References

- Hay P, Chinn D, Forbes D, Madden S, Newton R, Surgenor L, et al. Royal Australian and New Zealand College of Psychiatrists clinical practice guidelines for the treatment of eating disorders. Aust New Zealand J Psychiatry. 2014;48(11):977–1008. https://doi.org/10.1177/0004867414555814.
- Godart N, Perdereau F, Rein Z, Berthoz S, Wallier J, Jeammet P, et al. Comorbidity studies of eating disorders and mood disorders. Critical review of the literature. J Affect Disord. 2007;97(1–3):37–49. https://doi.org/10.1016/ j.jad.2006.06.023.
- Godart N, Flament M, Perdereau F, Jeammet P. Comorbidity between eating disorders and anxiety disorders: A review. Int J Eating Disord. 2002;32(3): 253–70. https://doi.org/10.1002/eat.10096.
- Haagsma JÅ, Graetz N, Bolliger I, Naghavi M, Higashi H, Mullany E, et al. The global burden of injury: incidence, mortality, disability-adjusted life years and time trends from the Global Burden of Disease study 2013. Injury Prevention. 2016;22:3–18. https://doi.org/10.1136/injuryprev-2015-041616.
- Butterfly Foundation. Paying the Price: The Economic and Social Impact of Eating Disorders in Australia. Sydney: Deloitte Access Economics; 2012.
- Hart L, Granillo M, Jorm A, Paxton S. Unmet need for treatment in the eating disorders: A systematic review of eating disorder specific treatment seeking among community cases. Clin Psychol Rev. 2011;31(5):727–35. https://doi.org/10.1016/j.cpr.2011.03.004.
- Steinhausen H. The Outcome of Anorexia Nervosa in the 20th Century. Am J Psychiatry. 2002;159(8):1284–93. https://doi.org/10.1176/appi.ajp.159.8.1284.
- Grange D, Loeb K. Early identification and treatment of eating disorders: prodrome to syndrome. Early Interv Psychiatry. 2007;1(1):27–39. https://doi. org/10.1111/j.1751-7893.2007.00007.x.
- Currin L, Schmidt U. A critical analysis of the utility of an early intervention approach in the eating disorders. J Mental Health. 2005;14(6):611–24. https://doi.org/10.1080/09638230500347939.
- Sim L, McAlpine D, Grothe K, Himes S, Cockerill R, Clark M. Identification and Treatment of Eating Disorders in the Primary Care Setting. Mayo Clinic Proc. 2010;85(8):746–51. https://doi.org/10.4065/mcp.2010.0070.
- Errichiello L, lodice D, Bruzzese D, Gherghi M, Senatore I. Prognostic factors and outcome in anorexia nervosa: a follow-up study. Eat Weight Disord. 2015;21(1):73–82. https://doi.org/10.1007/s40519-015-0211-2.
- Evans E, Hay P, Mond J, Paxton S, Quirk F, Rodgers B, et al. Barriers to Help-Seeking in Young Women with Eating Disorders: A Qualitative Exploration in a Longitudinal Community Survey. Eat Disord. 2011;19(3):270–85. https:// doi.org/10.1080/10640266.2011.566152.
- I thought I wasn't thin enough to be anorexic. BMJ. 2017;359;j5378. https:// doi.org/10.1136/bmj.j5378.
- Blake W, Turnbull S, Treasure J. Stages and processes of change in eating disorders: implications for therapy. Clin Psychol Psychother. 1997;4(3):186–91. https://doi.org/10.1002/(SICI)1099-0879(199709)4:3<186:AID-CPP128>3.0.CO;2-5.
- Musolino C, Warin M, Wade T, Gilchrist P. 'Healthy anorexia': The complexity of care in disordered eating. Soc Sci Med. 2015;139:18–25. https://doi.org/ 10.1016/j.socscimed.2015.06.030.
- 16. McDowell I. Measuring Health. New York: Oxford University Press; 2006.

- Lemouchoux C, Millar H, Naji S. Eating disorders in Scotland: starved of resources? Psychiatric Bulletin. 2001;25(7):256–60. https://doi.org/10.1192/pb. 25.7.256.
- Ivancic L, Maguire S, Miskovic-Wheatley J, Harrison C, Nassar N. (In press). Prevalence and management of people with eating disorders presenting to primary care: a national study. Aust New Zealand J Psychiatry. 2020.
- Fursland A, Watson H. Eating disorders: A hidden phenomenon in outpatient mental health? Int J Eat Disord. 2013;47(4):422–5. https://doi.org/ 10.1002/eat.22205.
- Fairburn CG, Beglin S. Eating disorder examination questionnaire (EDE-Q 6. 0). In: Fairburn CG, editor. Cognitive Behavior Therapy and Eating Disorders. New York: Guildford Press; 2008.
- Morgan J, Reid F, Lacey J. The SCOFF questionnaire: assessment of a new screening tool for eating disorders. BMJ. 1999;319(7223):1467–8. https://doi. org/10.1136/bmj.319.7223.1467.
- Cotton M, Ball C, Robinson P. Four simple questions can help screen for eating disorders. J General Internal Med. 2003;18(1):53–6. https://doi.org/10. 1046/j.1525-1497.2003.20374.x.
- Hill L, Reid F, Morgan J, Lacey JSCOFF. the development of an eating disorder screening questionnaire. Int J Eat Disord. 2009. https://doi.org/10. 1002/eat.20679.
- Mond J, Hay P, Rodgers B, Owen C. Self-report versus interview assessment of purging in a community sample of women. Eur Eat Disord Rev. 2007; 15(6):403 409. https://doi.org/10.1002/erv.792.
- Kagan S, Melrose C. The SCOFF questionnaire was less sensitive but more specific than the ESP for detecting eating disorders. Evid Based Nurs. 2003; 6(4):118.
- Gideon N, Hawkes N, Mond J, Saunders R, Tchanturia K, Development SL. Psychometric Validation of the EDE-QS, a 12 Item Short Form of the Eating Disorder Examination Questionnaire (EDE-Q). PLOS ONE. 2016;11(5): e0152744. https://doi.org/10.1371/journal.pone.0207256.
- Grilo C, Henderson K, Bell R, Crosby R. Eating Disorder Examination-Questionnaire Factor Structure and Construct Validity in Bariatric Surgery Candidates. Obes Surg. 2012;23(5):657–62. https://doi.org/10.1007/s11695-012-0840-8.
- Kliem S, Mößle T, Zenger M, Strauß B, Brähler E, Hilbert A. The eating disorder examination-questionnaire 8: A brief measure of eating disorder psychopathology (EDE Q8). Int J Eat Disord. 2015;49(6):613–6. https://doi. org/10.1002/eat.22487.
- Mond J, Hay P, Rodgers B, Owen C, Beumont P. Validity of the Eating Disorder Examination Questionnaire (EDE-Q) in screening for eating disorders in community samples. Behav Res Ther. 2004;42(5):551–67. https:// doi.org/10.1016/S0005-7967(03)00161-X.
- Graham A, Trockel M, Weisman H, Fitzsimmons-Craft E, Balantekin K, Wilfley D, et al. A screening tool for detecting eating disorder risk and diagnostic symptoms among college-age women. J Am Coll Health. 2018;67(4):357–66. https://doi.org/10.1080/07448481.2018.1483936.
- Fitzsimmons-Craft E, Balantekin K, Graham A, Smolar L, Park D, Mysko C, et al. Results of disseminating an online screen for eating disorders across the U.S.: Reach, respondent characteristics, and unmet treatment need. Int J Eat Disord. 2019;52(6):721–9. https://doi.org/10.1002/eat.23043.
- Schlachta-Fairchild L, Elfrink V. Models of Health Care and the Consumer Perspective of Telehealth in the Information Age. In: Nelson R, Ball M, editors. Consumer Informatics Health Informatics. New York: Springer; 2004.
- van der Eijk M, Nijhuis F, Faber M, Bloem B. Moving from physician-centered care towards patient-centered care for Parkinson's disease patients. Parkinsonism Relat Disord. 2013;19(11):923–7. https://doi.org/10.1016/j. parkreldis.2013.04.022.
- Lee K, Hoti K, Hughes J, Emmerton L. Dr Google Is Here to Stay but Health Care Professionals Are Still Valued: An Analysis of Health Care Consumers' Internet Navigation Support Preferences. J Med Int Res. 2017;19(6):e210. https://doi.org/10.2196/jmir.7489.
- Where are we now? | National Digital Health Strategy. Conversation. digitalhealth.gov.au. 2020. Available from: https://conversation.digitalhealth. gov.au/where-are-we-now. Cited 1 May 2020.
- Powell J, Inglis N, Ronnie J, Large S. The Characteristics and Motivations of Online Health Information Seekers: Cross-Sectional Survey and Qualitative Interview Study. J Med Int Res. 2011;13(1):e20. https://doi.org/10.2196/jmir.1600.
- Bashshur R, Doarn C, Frenk J, Kvedar J, Woolliscroft J. Telemedicine and the COVID 19 Pandemic, Lessons for the Future. Telemedicine e-Health. 2020; 26(5):571–3. https://doi.org/10.1089/tmj.2020.29040.rb.

- Wang C, Pan R, Wan X, Tan Y, Xu L, Ho C, et al. Immediate Psychological Responses and Associated Factors during the Initial Stage of the 2019 Coronavirus Disease (COVID-19) Epidemic among the General Population in China. Int J Environ Res Public Health. 2020;17(5):1729. https://doi.org/10. 3390/ijerph17051729.
- Torales J, O'Higgins M, Castaldelli-Maia J, Ventriglio A. The outbreak of COVID-19 coronavirus and its impact on global mental health. Int J Soc Psychiatry. 2020;66(4):317–20. https://doi.org/10.1177/0020764020915212.
- Bove C, Olson C. Obesity in low-income rural women: qualitative insights about physical activity and eating patterns. Women Health. 2006;44(1):57– 78. https://doi.org/10.1300/J013v44n01_04.
- Lydecker J, Grilo C. Food insecurity and bulimia nervosa in the United States. Int J Eat Disord. 2019. https://doi.org/10.1002/eat.23074.
- 42. Thornton J. Covid-19: A&E visits in England fall by 25% in week after lockdown. BMJ. 2020;369:m1401. https://doi.org/10.1136/bmj.m1401.
- Irvine, M. (2020). 4 Major trends caused by COVID-19 and how to respond. Retrieved from https://www.wordstream.com/blog/ws/2020/03/23/covid19. Businesstrends.
- 44. Are you at risk? InsideOut Institute for Eating Disorders (2018). Retrieved 22 February 2020, from https://insideoutinstitute.org.au/screener.
- Sheehan K. Crowdsourcing research: Data collection with Amazon's Mechanical Turk. Commun Monographs. 2017;85(1):140–56. https://doi.org/ 10.1080/03637751.2017.1342043.
- Garner D, Olmstead M, Polivy J. Development and validation of a multidimensional eating disorder inventory for anorexia nervosa and bulimia. Int J Eat Disord. 1983;2(2):15–34. https://doi.org/10.1002/1098-108X(198321)2:2<15::AID-EAT2260020203>3.0.CO;2-6.
- Garner D, Olmsted M, Bohr Y, Garfinkel P. The Eating Attitudes Test: psychometric features and clinical correlates. Psychol Med. 1982;12(4):871–8.
- Berg K, Peterson C, Frazier P, Crow S. Psychometric evaluation of the eating disorder examination and eating disorder examination-questionnaire: A systematic review of the literature. Int J Eat Disord. 2011;45(3):428–38. https://doi.org/10.1002/eat.20931.
- Chan C, Leung S. Validation of the Eating Disorder Examination Questionnaire: an online version. J Human Nutr Dietetics. 2014;28(6):659–65. https://doi.org/10.1111/jhn.12275.
- Mond J, Myers T, Crosby R, Hay P, Rodgers B, Morgan J, et al. Screening for eating disorders in primary care: EDE-Q versus SCOFF. Behav Res Ther. 2008; 46(5):612–22. https://doi.org/10.1016/j.brat.2008.02.003.
- Crowne D, Marlowe D. A new scale of social desirability independent of psychopathology. J Consult Psychol. 1960;24(4):349–54. https://doi.org/10. 1037/h0047358.
- Barger S. The Marlowe-Crowne Affair: Short Forms, Psychometric Structure, and Social Desirability. J Personal Assess. 2002;79(2):286–305. https://doi.org/ 10.1207/S15327752JPA7902_11.
- Tanaka-Matsumi J, Kameoka V. Reliabilities and concurrent validities of popular self report measures of depression, anxiety, and social desirability. J Consult Clin Psychol. 1986;54(3):328–33. https://doi.org/10.1037/0022-006X. 54.3.328.
- Baxter S, Smith A, Litaker M, Baglio M, Guinn C, Shaffer N. Children's Social Desirability and Dietary Reports. J Nutr Educ Behav. 2004;36(2):84–9. https:// doi.org/10.1016/S1499-4046(06)60138-3.
- Crandall V, Crandall V, Katkovsky W. A children's social desirability questionnaire. J Consult Psychol. 1965;29(1):27–36. https://doi.org/10.1037/ h0020966.
- Miller P, Baxter S, Hitchcock D, Royer J, Smith A, Guinn C. Test-Retest Reliability of a Short Form of the Children's Social Desirability Scale for Nutrition and Health-related Research. J Nutr Educ Behav. 2014;46(5):423–8. https://doi.org/10.1016/j.jneb.2013.11.002.
- 57. PASS 15 Power Analysis and Sample Size Software (2017). NCSS, LLC. Kaysville, Utah, USA, ncss.com/software/pass.
- Bolarinwa O. Principles and methods of validity and reliability testing of questionnaires used in social and health science researches. Nigerian Postgraduate Med J. 2015;22(4):195. https://doi.org/10.4103/1117-1936.173959.
- A Step-by-Step Approach to Using SAS[®] for Univariate & Multivariate Statistics. JMP[®] for Basic Univariate and Multivariate Statistics. Technometrics. 2006;48(2):314-314.
- Alfadl AA, Ibrahim MI, Hassali MA. Scale development on consumer behavior toward counterfeit drugs in a developing country: a quantitative study exploiting the tools of an evolving paradigm. BMC Public Health. 2013;13(1). https://doi.org/10.1186/1471-2458-13-829.

- 61. Nunnally J. Psychometric theory. New York: McGraw-Hill Book; 1978.
- Koo T, Li M. A Guideline of Selecting and Reporting Intraclass Correlation Coefficients for Reliability Research. J Chiropractic Med. 2016;15(2):155–63. https://doi.org/10.1016/j.jcm.2016.02.012.
- Anastasi A. Psychological Testing. 7th ed. London: Simon & Schuster; 1997. p. 118–9.
- Parikh R, Mathai A, Parikh S, Chandra Sekhar G, Thomas R. Understanding and using sensitivity, specificity and predictive values. Indian J Ophthalmol. 2008;56(1):45. https://doi.org/10.4103/0301-4738.37595.
- Florkowski CM. Sensitivity, specificity, receiver-operating characteristic (ROC) curves and likelihood ratios: communicating the performance of diagnostic tests. Clin Biochem Rev. 2008;29(Suppl 1):S83–7.
- Anastasi A. Psychological Testing. 7th ed. London: Simon & Schuster; 1997. p. 91–102.
- Akoglu H. User's guide to correlation coefficients. Turkish J Emerg Med. 2018;18(3):91–3. https://doi.org/10.1016/j.tjem.2018.08.001.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

